1. Reply Reviewer #1

We have made overall revisions to the manuscript in accordance with the format required by the journal and edited the language accordingly.

2. Reply Reviewer #2

(1) We have made overall revisions to the manuscript in accordance with the format required by the journal and edited the language accordingly.

(2) We have rewritten the background according to the journal format requirements and added aim.

(3) We have increased the resolution of the image to 600 to make it clearer.